Author's response to reviews

Title: Symptoms of depression impact the course of lung function in adolescents and adults with cystic fibrosis

Authors:

Astrid Fidika (astrid.fidika@uniklinik-ulm.de)
Marion Herle (marion.herle@uniklinik-ulm.de)
Lutz Goldbeck (lutz.goldbeck@uniklinik-ulm.de)

Version: 2
Date: 12 August 2014

Author's response to reviews:

please see over